

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Kilponen, Juha et al.

Working Paper

Comparing fiscal multipliers across models and countries in Europe

NBB Working Paper, No. 278

Provided in Cooperation with:

National Bank of Belgium, Brussels

Suggested Citation: Kilponen, Juha et al. (2015): Comparing fiscal multipliers across models and countries in Europe, NBB Working Paper, No. 278, National Bank of Belgium, Brussels

This Version is available at: https://hdl.handle.net/10419/144490

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



Comparing fiscal multipliers across models and countries in Europe



by Juha Kilponen, Massimiliano Pisani, Sebastian Schmidt, Vesna Corbo, Tibor Hledik, Josef Hollmayr, Samuel Hurtado, Paulo Júlio, Dmitry Kulikov, Matthieu Lemoine, Matija Lozej, Henrik Lundvall, José R. Maria, Brian Micallef, Dimitris Papageorgiou, Jakub Rysanek, Dimitrios Sideris, Carlos Thomas and Gregory de Walque

March 2015 No 278



Editorial Director

Jan Smets, Member of the Board of Directors of the National Bank of Belgium

Statement of purpose:

The purpose of these working papers is to promote the circulation of research results (Research Series) and analytical studies (Documents Series) made within the National Bank of Belgium or presented by external economists in seminars, conferences and conventions organised by the Bank. The aim is therefore to provide a platform for discussion. The opinions expressed are strictly those of the authors and do not necessarily reflect the views of the National Bank of Belgium.

Orders

For orders and information on subscriptions and reductions: National Bank of Belgium, Documentation - Publications service, boulevard de Berlaimont 14, 1000 Brussels

Tel +32 2 221 20 33 - Fax +32 2 21 30 42

The Working Papers are available on the website of the Bank: http://www.nbb.be

© National Bank of Belgium, Brussels

All rights reserved.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

ISSN: 1375-680X (print) ISSN: 1784-2476 (online)

Abstract

This paper employs fifteen dynamic macroeconomic models maintained within the European System of Central Banks to assess the size of fiscal multipliers in European countries. Using a set of common simulations, we consider transitory and permanent shocks to government expenditures and different taxes. We investigate how the baseline multipliers change when monetary policy is transitorily constrained by the zero nominal interest rate bound, certain crisis-related structural features of the economy such as the share of liquidity-constrained households change, and the endogenous fiscal rule that ensures fiscal sustainability in the long run is specified in terms of labour income taxes instead of lump-sum taxes.

JEL classification: E12, E13, E17, E62, E63.

Keywords: Fiscal policy, Output multipliers, Model comparison, Zero lower bound.

Corresponding author:

Gregory de Walque, Research Department, NBB - e-mail: gregory.dewalque@nbb.be

The authors thank Günter Coenen and the members of the Working Group on Econometric Modelling for helpful comments.

The views expressed in this paper are those of the authors and do not necessarily reflect the views of the National Bank of Belgium or any other institution to which the authors are affiliated.

TABLE OF CONTENTS

I.	Introduction	1
II.	Model set-up	3
	a. General features	3
	b. Steady state values and calibration	4
III.	Simulation experiments and results	
	a. Definition of fiscal multipliers and simulated scenarios	
	b. Fiscal multipliers for temporary fiscal shocks	
	c. Fiscal multipliers for permanent fiscal shocks	13
IV.	Sensitivity analysis	18
٧.	Comparison of model-based fiscal multipliers and empirical results	20
VI.	Conclusions	22
Refe	rences	24
Table	es	29
Appe	endix	39
Natio	onal Bank of Belgium - Working papers series	45

I. Introduction

Fiscal multipliers vary in many dimensions, as highlighted by the recent academic literature. The disagreement over the size of fiscal multipliers over the last years has not yet been resolved despite voluminous empirical and theoretical analyses of the issue.

This article provides a quantitative assessment of the key factors that determine the GDP effects associated with the use of alternative fiscal instruments. We employ structural macroeconomic models maintained within the European System of Central Banks (ESCB) to document by means of simulations how multipliers depend on the fiscal instruments, the zero lower bound (ZLB) on the monetary policy rate, the duration of the fiscal shock and various country-specific features. In each of the simulated scenarios, we consider the short and – if applicable – the long-run effects of a discretionary change in a single fiscal policy instrument on real GDP. The change in the policy instrument amounts to 1% of baseline GDP and represents a tightening of the fiscal stance. Specifically, we consider a reduction in (unproductive) government consumption and increases in the households' labour income tax rate, the capital income tax rate and the consumption tax rate.

The multipliers are computed assuming that monetary policy is either determined by an endogenous Taylor-type nominal interest rate rule, or alternatively, that interest rates are fixed for a period of two years, which we take as mimicking a binding ZLB condition.

The key findings from our simulations can be summarized as follows. First, short-run multipliers are in general negative and smaller than one in absolute value. This result is quite robust with respect to the fiscal instrument, the considered country and the duration of the fiscal shock. Short-run tax (labour, consumption and capital) multipliers are typically smaller in absolute value than government consumption multipliers.

Second, imposing the ZLB to bind for two years does not greatly affect shortrun multipliers associated with a temporary fiscal tightening of individual euro area countries. The reason is that the monetary policy rate stays essentially at its baseline level even when the monetary authority is free to adjust it, reflecting the limited impact of a country-specific fiscal shock on the euro area economy. In contrast, the ZLB unfolds quite sizeable effects on the size of multipliers if the fiscal shocks are simultaneously implemented in the euro area as a whole. In particular, short-run government consumption multipliers become larger than one. The same holds true for non-euro area countries in which monetary policy is determined domestically.

Third, the short and the long-run effects of permanent fiscal shocks depend on the fiscal instrument that reacts endogenously to stabilise public debt in the long run. Long-run multipliers are in general negative when the budgetary room materialising after the fiscal tightening is used to reduce lump-sum taxes. Instead, long-run multipliers are typically positive if the households' labour income tax rate is reduced in the medium to long term. Since households anticipate these long-run GDP effects at the outset of the simulations, short-run multipliers tend to be more favourable if distortionary taxes are used to stabilise public debt.

Our paper is related to a small set of studies that examine the robustness of fiscal multiplier estimates across structural models. Cwik and Wieland (2011) use five macroeconomic models to estimate multipliers associated with the European Economic Recovery Plan and related national fiscal policy measures in the euro area. They focus on the announced government purchases component of the plan for 2009 and 2010. In the majority of models, private consumption and investment are crowded out by the rise in government spending unless the zero nominal interest rate bound is anticipated to be binding for at least two years. Unlike our paper, they do not consider tax policies. Coenen at al. (2012) employ seven dynamic stochastic general equilibrium models maintained by policymaking institutions to assess the GDP effects of discretionary fiscal stimulus shocks to seven different instruments. They find that fiscal stimulus is most effective if it is temporary and accompanied by an accommodative monetary policy stance. Unlike our paper, they do not focus on European countries. We should also emphasise that both studies investigate multipliers associated with expansionary fiscal

shocks whereas we consider fiscal retrenchments. The sign of the fiscal shocks matter in particular in those situations where the economy is at the zero lower bound.

More broadly, our paper is related to a large and growing set of studies that examine the size of fiscal multipliers within one or two macroeconomic models. Prominent recent examples include Cogan et al. (2010), Christiano et al. (2011), Eggertsson (2011) and Woodford (2011).

The remainder of the paper is structured as follows. Section II summarizes the models used in the simulation exercises. Section III describes the standardised simulations and presents the results. Section IV summarizes the results from the sensitivity analysis. Section V compares the model-based results with standard fiscal multiplier estimates from the empirical literature. Finally, Section VI concludes.

II. Model set-up

a. General features

We use fifteen models from National Central Banks (NCBs) and the ECB in the simulation exercises. Fourteen out of fifteen are New-Keynesian dynamic general equilibrium models, ten out of fifteen are calibrated. A complete list of the models is presented in Table A1 in the Appendix.

The majority of models from NCBs of the euro area are based on multi-country set-ups, namely those of Belgium, Estonia, France, Germany, Italy, Malta, Slovenia and Spain. These models exhibit a "home" country, the rest of the euro area (possibly subdivided) and in some cases the rest of the world. In these models the euro area monetary policy responds to economic fluctuations in the home country only proportionally to its weight in the monetary union.

A second set of models comprises small open economy set-ups, with an exogenous rest of the euro area and/or rest of the world: Czech Republic,

Finland, Greece, Netherlands, Portugal and Sweden. If the corresponding country is part of the euro area, monetary policy is assumed to be exogenous.

Finally, the ECB's New Area-Wide Model (NAWM) has also been used. It is a two-country model of the euro area and the United States. Monetary policy in both model blocks is characterized by standard nominal interest rate rules.

Responses to fiscal shocks can be influenced by the fiscal instrument that, through the fiscal rule, endogenously adjusts to stabilise public debt. In the vast majority of the models, this fiscal instrument reacts to deviations of the government debt-to-GDP ratio from the target, but in a few cases the fiscal instrument reacts also to deviations of the public deficit or public consumption from its long-run target. Typically, either the labour income tax or lump-sum transfers are used as fiscal instrument. In some of the simulations, the choice of the fiscal rule has been left at the discretion of each country modelers. However, whenever the fiscal rule becomes critical for the results, we harmonized the instrument that is specified by the rule across models.

b. Steady state values and calibration

Key parameters and their calibration are listed in Tables A2-A4 in the Appendix. The models differ in various aspects.

In terms of steady state values, the models differ significantly as regards the imports-to-GDP ratio, which to some extent measures the degree of openness of the economy. Lowest import penetration is found for Greece and largest for Estonia. The models differ also substantially in terms of how public expenditures are financed. As an example, in the German model the labour income tax revenues amount to 35% of GDP, while in Spain they account only for 7% of GDP. The steady state values of debt-to-GDP ratio vary from 0% to 120%. The models also vary by the degree of home bias in government consumption. Most of the models assume full home bias, as is typical in this type of set-ups, and only a few feature somewhat lower home bias of around 90%. Finally, the share of liquidity-constrained consumers, i.e. agents that

have at most limited ability to smooth consumption over time, varies between 0 to 40%.

Regarding the calibration of some key parameters, household preferences, investment (or capital) adjustment costs, price and wage stickiness (characterized by a Calvo parameter or, if the value is larger than 1, by a Rotemberg adjustment cost parameter) and the proportion of firms (workers) that index their price (wage) to inflation are quite different across models: The Frisch elasticity of labour supply varies from 0.50 to 11, wage indexation from 0 to 0.90, and investment adjustment costs from 0.20 to about 14.

All these differences can play an important role in explaining differences in fiscal multipliers across the models.

III. Simulation experiments and results

a. Definition of fiscal multipliers and simulated scenarios

In each of the simulation scenarios reported below, we consider the short and – if applicable – the long-run effects of a discretionary change in a single fiscal policy instrument on real GDP. The change in the policy instrument amounts to 1% of baseline (before shock) GDP and represents a tightening of the fiscal stance. Specifically, we consider a reduction in government consumption and increases in the labour income tax rate of households, the capital income tax rate and the consumption tax rate. Fiscal instruments other than the ones subject to discretionary change are held constant for the first two years. Also the social security contributions are held unchanged in the simulations. In the medium to long run, either lump-sum or labour income taxes are allowed to

¹ In the case of the Swedish model, the fiscal rule is implicit: lump-sum transfers make sure that government expenditures and tax revenues are equal in every period. For the simulations carried out in this model, lump-sum transfers are thus allowed to adjust also in the short run.

adjust according to the country-specific fiscal rules to stabilise the public debt-to-GDP or deficit-to-GDP ratio at their target values. In the case of permanent fiscal shocks, the multipliers can be quite sensitive to the fiscal instrument that stabilises the debt or the deficit. Therefore, we conduct these simulations twice with each model, in one case imposing a lump-sum tax rule and in the other a households' labour income tax rule.²

Monetary policy is harmonized across models, assuming that the short-term nominal interest rate is determined by the Taylor rule used in Gomes et al. (2012), where the policy rate responds to euro area-wide inflation and output growth.³ We also assess the role of the ZLB on the monetary policy rate for the fiscal multipliers, assuming that the Taylor rule is deactivated and the short-term nominal interest rate is held constant at its baseline level for 2 years.

All simulations are run under perfect foresight. Therefore, policies are fully anticipated by households and firms.

We first report GDP multipliers for transitory changes in each fiscal instrument implemented unilaterally by a single country. Subsequently we present multipliers associated with permanent changes in each fiscal instrument. In both cases, two years after the initial shock the country-specific fiscal rule starts to operate, slowly bringing the debt-to-GDP ratio or deficit-to-GDP ratio back to its target level (the initial pre-shock level).

_

² The specification of the country-specific fiscal rule has only very modest effects on multipliers if the fiscal shock is transitory.

³ The rule is specified as $R_t = \rho R_{t-1} + (1-\rho)[\bar{R} + \theta_\pi (\pi_t - \bar{\pi})] + \theta_y Y g r_t$, where R denotes the policy rate, π is the annual euro area inflation rate (excluding the direct effect from changes in consumption taxes) and Y g r denotes quarterly euro area output growth. \bar{R} is the equilibrium nominal interest rate and $\bar{\pi}$ is the monetary authority's inflation target. The coefficients are as follows: $\rho = 0.87$, $\bar{\pi} = 1.02$, $\theta_\pi = 1.70$ and $\theta_y = 0.10$.

b. Fiscal multipliers for temporary fiscal shocks

Government consumption

Table 1 shows the government consumption multipliers, i.e. the response of GDP to a temporary change in government consumption. The latter, as a ratio to the baseline GDP, decreases by one percentage point for two years and then returns to the baseline.

In the first case, denoted 'No ZLB', the ZLB is not imposed as a constraint and the nominal interest rate adjusts according to the Taylor rule (see footnote 9). In the second case, denoted '2 year-ZLB', the nominal interest rate is kept constant during the first two years of the simulation and follows the Taylor rule thereafter. Similarly, all the other fiscal instruments are held constant for two years. The fiscal rule kicks in thereafter.

- Table 1 around here -

When the ZLB is not binding all multipliers are below one in absolute terms. In the majority of the models the first year multipliers are between 0.7 and 0.9, but in some cases they are lower. They are close to 0.5 in Germany, Spain, Czech Republic and Sweden.

Multipliers are lower than one in absolute value because the crowding-in effect on private sector spending partially compensates for the reduction in public consumption. In the majority of models, private sector consumption and investment (not reported) increase, as lower public consumption increases the resources available for the private sector. However, the positive wealth effect, i.e. the reduction in the present value of future tax payments required to balance the government's budget, is rather small, as the fiscal retrenchment is temporary.

Multipliers are lower in absolute terms in the second year than in the first year. The presence of adjustment costs on investment, of habit persistence in

consumption and of nominal wage and price rigidities makes the response of private spending gradual. For instance, the difference in first and second year multipliers is large in the case of Portugal, which features relatively large investment adjustment costs.

Note that in the case of euro area countries the country-specific real interest rate does not greatly decrease and, hence, does not contribute significantly to the crowding-in of private demand. There are two reasons for this. First, the monetary policy rate, set at the union-wide level, is not greatly reduced after a shock, because each individual country has a small effect on euro area inflation and economic activity. Second, the response of the country-specific inflation rate is rather contained. In the case of Portugal, the multipliers increase around 10% in the first year when the model includes financial frictions. Along with lower aggregate demand, the price of capital decreases, as well as net worth. The entrepreneurial sector becomes more leveraged and is forced to face a higher external finance premium, which dampens investment. The presence of financial frictions also creates some persistence effects, as it takes time to rebuild lost net worth.

The zero lower bound

For euro area countries, multipliers are lower than one also under the ZLB (the only exception is France, which exhibits a multiplier slightly larger than one). For euro area countries other than Germany, multipliers are either unchanged or only slightly larger than in the case of the nominal interest rate reacting according to the Taylor rule. The ZLB does not greatly affect economic activity in the euro area, because it lasts for a relatively small (but plausible) number of periods (8 quarters) and the country-specific fiscal shock produces only small cross-country spillovers. Responses of economic activity and inflation in the rest of the euro area under the ZLB are muted, as in the case of the active Taylor rule. As such, the responses of the region-specific real interest rates (in the considered country and in the rest of the euro area) are muted and similar in both scenarios.

However, the ZLB leads to a significantly larger multiplier in the case of a reduction of public consumption at the euro area level, as reflected in the simulation results from the ECB's NAWM. When the ZLB does not bind, the euro area policy rate is reduced, favouring the crowding-in of private spending. When it is binding, the constant nominal interest rate and the decrease in euro area inflation lead to a rather strong increase in the euro area real interest rate that depresses private spending. For similar reasons, multipliers become significantly larger when the ZLB binds in the case of Czech Republic and Sweden, reaching values that are clearly larger than one.

Different taxes

Households' labour income tax rate

Table 2 presents the short-run GDP multipliers in case of a transitory (two-year) increase in the households' labour income tax rate.

Multipliers are lower than one in absolute value and smaller than those associated with the reduction in government consumption. They are generally around 0.1 in the first year and between 0.2-0.4 in the second year. The increase in labour taxes has a small impact on GDP as it operates mainly through its effects on wealth (permanent income) and incentives to work. As in the case of public consumption, the wealth effect is rather small because the fiscal measure is transitory. Public consumption has a larger multiplier because it directly reduces aggregate demand.

The impact of the labour income tax increase is larger in some of those models that feature strong non-Ricardian features, including a relatively high percentage of liquidity-constrained consumers, such as in case of Greece, Portugal and Czech Republic. To some extent, the size of the labour tax multipliers is also related to the share of labour income tax revenues-to-GDP and to the degree of wage indexation. For countries with a large labour income tax base, the multiplier tends to be smaller in absolute terms. This is explained by the fact that e.g. labour supply reacts to a change in the labour

income *tax rate*, whereby a change in the tax rate needs to be smaller for those countries with a large labour income tax base to achieve an 1% increase in the ratio of labour income tax revenues to GDP. Stronger wage indexation, in turn, makes wages more backward looking and therefore limits the temporary increase in wages due to the tax hike. This limits the short-run impact of the temporary tax hike on employment and economic activity, resulting in a smaller multiplier.

- Table 2 around here -

Unlike in the scenario prescribing a reduction in government consumption, a labour tax hike leads to an increase in the multiplier for the majority of countries when moving from the first to the second year. This reflects the presence of nominal and real frictions, which leads to a gradual response of aggregate demand to the labour income tax hike through the substitution effect.

The labour income tax hike scenario also differs from the government consumption-based scenario in that for the majority of countries the multipliers are slightly smaller when the nominal interest rate is held constant for two years. The contained increase in inflation (associated with negative supply side effects of higher labour taxes) and the fixed policy rate assumption together result in a slight decrease in the real interest rate, partially limiting the decrease in aggregate demand. One exception to the general result is Germany, whose multipliers are slightly larger when the monetary policy rate is constant. In the Portuguese case, the presence of credit market frictions does not change the multipliers as the effect on prices, particularly the price of capital, is rather muted.

For the ZLB experiments in the Swedish model, the effects of the shocks to the fiscal policy instruments on the interest rate have been offset by anticipated monetary policy shocks and this may explain why the multipliers change signs under the ZLB. As discussed by Laséen and Svensson (2011) and by Carlstrom, Fuerst and Paustian (2012), anticipated monetary policy shocks may under some circumstances have implausibly large effects on inflation and output. The results from the ZLB experiments with the Swedish model should therefore be interpreted with some caution.

Capital income tax rate

Table 3 shows the short-run output multipliers of a transitory (two-year) increase in capital income taxation. The multipliers are generally rather small, below 0.3 in absolute terms. The short-run response of investment to an increase in the capital income tax is rather gradual, because of the short-run adjustment costs of investment. Moreover, there is no strong incentive to reduce investment since the increase in the capital income tax is transitory.

There are some exceptions. Multipliers are rather large in the case of Sweden and Greece. In the Greek model, the large multiplier is driven by the sizable reduction in the utilisation rate of capital and the price of capital that induce a strong negative response of output to the tax shock. In the Portuguese case, credit market frictions work to propagate and amplify the negative impact on GDP. Increasing capital income taxes directly affects entrepreneurial returns, increases leverage and the cost of external finance, which reduces investment.

Multipliers increase slightly under the 2-year ZLB scenario. As in the previous simulations, the decrease in union-wide inflation and economic activity due to the temporary drop in the country-specific demand is rather muted for countries belonging to the euro area. Under standard monetary policy, the policy rate does not greatly change and the country-specific real interest rate hardly moves. Similarly, the slowdown in country-specific inflation, and hence the increase in the country-specific real interest rate, is small under the ZLB. As a result, the ZLB does not significantly amplify the negative macroeconomic effects of the capital income tax increase. Concerning the effects for Sweden, the results under the ZLB should again be interpreted with some caution (see the discussion of the labour income tax hike simulations under the ZLB). In the Portuguese case, the presence of credit market

frictions has a slight amplification effect on the multipliers, also creating some persistence effects.

- Table 3 around here -

Consumption tax rate

Table 4 shows the short-run output multipliers associated with a transitory increase in consumption taxation. In the absence of the ZLB, all multipliers are below one in absolute value. The largest multiplier is equal to 0.7 and the smallest is equal to 0.1. These differences reflect, at least partly, differences in the calibration of the inter-temporal elasticity of substitution and consumption habit persistence. A higher inter-temporal elasticity of substitution and lower habit persistence make current consumption more responsive to changes in consumer prices which are directly affected by the transitory increase in consumption taxes. Habit persistence also tends to increase the multiplier in the second year relative to the first year, because households favour a gradual response of private consumption.

- Table 4 around here -

The ZLB does not change the overall picture significantly. The only exceptions are the euro area and Sweden. In the case of the euro area, where private consumption accounts for about 60% of GDP, a decrease in euro area consumption reduces inflation and, hence, the increase in the euro area real interest rate is relatively strong under the ZLB. For Sweden this effect is even stronger, leading to a multiplier larger than one.

c. Fiscal multipliers for permanent fiscal shocks

In the previous section, we have considered transitory fiscal shocks. We now turn to permanent fiscal shocks, which allow us to assess not only short but also long-run effects of discretionary changes in fiscal instruments. A permanent fiscal shock can be interpreted as "fiscal reform", which permanently alters the fiscal structure of the economy. For instance, a permanent reduction in government consumption, associated with the reduction in labour income taxes, reduces the size of the public sector and tax burden of the economy permanently. Similarly, a permanent change in one type of tax financed by another type of tax

represents a permanent change in the tax structure of the economy. Following the previous simulations, the fiscal rule is deactivated for the first two years. Thereafter it becomes active again and it stabilises the public debt and/or the deficit at their target values, which remain unchanged.

Since the long-run response of output critically depends on the fiscal instrument that is determined by the fiscal rule, we compare two cases. In the first case, the fiscal rule is specified in terms of the lump-sum tax, representing the benchmark assumption typically employed in the DSGE literature.⁴ In the second, arguably more plausible case, the fiscal rule is instead specified in terms of the (more distortionary) households' labour income tax. It turns out that the choice of the instrument that serves to ensure fiscal sustainability is not innocuous, especially as regards the long-run multipliers.

Government consumption, fiscal rule in terms of lump-sum taxes

The first three columns of Table 5 show the short and long-run output multipliers of a permanent reduction in government consumption when the fiscal rule is specified in terms of the lump-sum tax.

⁴ See, for instance, Christiano, Eichenbaum and Rebelo (2011).

The estimated short-run multipliers are smaller than one in absolute value, ranging from 0.25 to 0.97 in the first year. The multipliers are generally smaller than their counterparts in the case of a transitory reduction in public consumption (see Table 1), because of the large positive wealth effect on households and firms. The permanent reduction in public consumption makes resources available for private spending on a permanent basis; this induces a larger crowding-in effect on private consumption and investment. As in the case of transitory shocks, the multipliers are smaller in the second than in the first year, because nominal and real rigidities lead to a gradual adjustment of private demand for consumption and investment.

The long-run multipliers are negative across all models and, with the exception of the Greek model, remain smaller than one in absolute value. In the long run, a decrease in government consumption translates into lower lump sum taxes for households. Since lump sum taxes or transfers do not alter labour supply of Ricardian households or impact on relative prices in the long run, lower aggregate demand due to lower public expenditures lead to a negative GDP effect.⁵

Government consumption, fiscal rule in terms of labour taxes

The last three columns of Table 5 show the short and long-run multipliers of a permanent reduction in government consumption when the fiscal rule is specified in terms of the households' labour income tax rate. Short-run multipliers are generally smaller than in the case of the lump-sum tax rule. The anticipation of lower future labour income taxes induces households to gradually increase their labour effort. The anticipation of higher labour income in the medium and long run strengthens the crowding-in effect on private demand already in the short run, leading to a lower short-run GDP multiplier.

In contrast to the previous results, long-run multipliers are now positive and, in some cases, larger than one. Typically, multipliers turn positive after three to

_

⁵ This result is in line with earlier work by the WGEM, see ECB (2011).

six years. The permanent reduction in the labour tax rate leads to an outward shift of labour supply, providing incentives to increase employment. Higher employment in turn makes capital more productive. Since capital is rather elastic in the long run, there is a relatively large (supply-side) effect on production and economic activity. The largest multiplier is equal to 1.6 (Portuguese model), the smallest to 0.1 (German model). Interestingly, the long-run multipliers tend to be smaller and hence economic benefits of the reform are smaller for those countries that have a higher import penetration, i.e. higher import-to-GDP ratio. Larger trade linkages imply that a larger share of aggregate demand is satisfied by imports.

- Table 5 around here -

Distortionary taxes, fiscal rule in terms of lump-sum taxes

Table 6 shows the multipliers associated with a permanent increase in different tax revenues when the fiscal rule is specified in terms of lump-sum taxes.

We first consider the permanent increase in labour income taxes. Short-run multipliers are negative and generally lower than one in absolute value, ranging between 0.0 and 0.8 in the first year and between 0.1 and 1.0 in the second year. Long-run multipliers are negative as well and in the majority of cases larger than one in absolute value, reflecting the distortionary nature of labour income taxation. Due to the negative long-run GDP response and the associated wealth effect, short-run multipliers are in general larger than in the case of a transitory fiscal shock (see Table 2). The estimates of short-run multipliers associated with capital income taxation vary quite a lot across

⁶ This difference partly reflects the fact that fiscal rules for some models are different under transitory and permanent shocks, but the general conclusion about the negative wealth effect still holds.

models. In absolute values, the range goes from 0.0 for the German model in the first year to 2.5 in case of the Greek model in the second year. Long-run multipliers are unequivocally negative and much larger in absolute value than the multipliers associated with labour taxation. Physical capital fully adjusts to the new tax level, inducing a strong decline in labour and hence economic activity. Long-run multipliers are larger than 3 in France, Greece, Slovenia and Spain, and equal to or larger than 2 in the euro area, Belgium, Finland, Italy and Portugal. Both short-run and long-run multipliers tend to be larger (in absolute terms) for those countries in which the private investment-to-GDP ratio is larger and where the initial capital tax revenues are lower. In the Portuguese model, financial frictions amplify the negative short-run impact on GDP, as a deterioration of entrepreneurs' net worth, due to higher capital income taxes, increases leverage and the cost of external funds (a result already highlighted in section IV b).

Table 6 around here –

Finally, the short-run multipliers associated with the consumption tax hike are between 0.0 and 0.5 in absolute values, while long-run multipliers are between 0.3 and 0.7 (the only exceptions are Greece, whose long-run multiplier is about one, and Spain, whose long-run multiplier is 0.0). They are larger than the corresponding short-run multipliers because consumption habits lead to a gradual response of consumption to the increase in taxation. For the same reason the multiplier is usually larger in the second than in the first year.

Distortionary taxes, fiscal rule in terms of labour taxes

16

_

⁷ Only Germany has a multiplier lower than one.

Finally, we assess the value of multipliers when the fiscal room created by the permanent increase in capital income or consumption taxation is used to permanently reduce households' labour income taxes (instead of lump-sum taxes). Table 7 reports the results.

Short-run multipliers associated with a permanent increase in capital income taxes are somewhat smaller in absolute values than in the case where the fiscal rules are specified in terms of the lump-sum tax. In the case of Finland, the multiplier even becomes positive in the first year. The anticipation of the permanent reduction in labour taxation provides an incentive to gradually increase labour supply. This partially counterbalances the incentive to reduce investment associated with a higher taxation of capital. Long-run multipliers are again much larger than one in absolute value in most cases, given that investment is very elastic in the long run. As such, the expansionary effects of lower labour taxation compensate only partially the strong recessionary effect of permanently higher capital income taxes.

Short-run multipliers associated with a permanent increase in consumption taxes are lower when the fiscal rule ensuring government debt sustainability is specified in terms of the labour income tax instead of the lump-sum tax, and in some cases even become positive (Estonia, Italy, Slovenia) due to the quick positive response of labour and the gradual response of consumption. The permanent reduction in labour taxes partially compensates for the permanent increase in consumption taxation. In the case of Estonia, the rather large multiplier is explained by strong competitiveness gains due to reduced labour costs, combined with the fact that trade effects in the Estonian model have a relatively large weight in the overall dynamics.

- Table 7 around here -

In contrast to the capital income tax-based scenario, the long-run consumption tax-based multipliers are positive. Lower labour taxes favour the

increase in employment, counterbalancing the negative effects due to an increase in consumption taxes. As such, economic activity increases in the long run. Higher short-run consumption tax multipliers tend to be associated with models that exhibit a larger share of liquidity-constrained consumers and a higher coefficient of risk aversion. Liquidity-constrained consumers have at most a limited ability to smooth consumption over time, hence their reaction to the consumption tax hike is large.

IV. Sensitivity analysis

Results presented so far are based on the benchmark calibrations of the models as described in Tables A2-A4 in the Appendix. In this section, we analyse the sensitivity of the results with respect to the following changes in the models' calibration:

- 30 percentage point increase in the share of liquidity-constrained households
- 10% reduction in the degree of price stickiness
- 10% reduction in the degree of wage stickiness
- 50% reduction in households' risk aversion
- 50% increase in investment adjustment costs

The sensitivity analysis is conducted with and without the ZLB. It focuses on two scenarios, a permanent reduction in government consumption and a permanent increase in labour income taxes. All other fiscal instruments are held constant for the first two years. After two years, lump-sum taxes are allowed to adjust according to the fiscal rules. Only a subset of models was used (euro area, Finland, Italy, Malta, Portugal, and Slovenia). Table A5 in the Appendix reports the average short-run and long-run multipliers across models.

The sensitivity analysis shows that short-run multipliers become larger in absolute terms when there are more liquidity-constrained households,

reflecting the fact that these households are less able to smooth consumption than unconstrained households. This effect becomes exacerbated when the ZLB is binding. Absent the ZLB, the short-run government consumption multipliers are typically smaller when prices are less sticky. Firms adjust goods prices faster, leading to a quicker accommodating monetary policy response. Results are similar with regard to wage stickiness. Absent the ZLB, the short-run government consumption multipliers are typically smaller when wages are more flexible. A lower degree of risk aversion translates into a higher interest rate elasticity of aggregate demand so that the accommodating monetary policy response has a stronger effect, thereby lowering short-run spending multipliers. It should also be noted that the fiscal multipliers are sensitive to the degree of financial frictions as shown in Tables 1-9 for the Portuguese model. The presence of financial frictions increases in particular the government consumption and the capital income tax multipliers. For other taxes, however, these frictions seem less relevant.

At the same time, the sensitivity of the multipliers with respect to investment adjustment costs differs across models, thus precluding the derivation of any straightforward conclusion.

As for the permanent increase in households' labour taxes, we again find that, typically, having more liquidity-constrained households leads to a stronger GDP effect in the short run. Again, the sensitivity of the short-run multipliers to the share of liquidity-constrained households gets amplified when the ZLB is binding (in those models that exhibit an endogenous monetary policy rule). Another relatively robust finding is that, contrary to the permanent spending shocks, lower risk aversion leads to stronger output effects, and hence higher multipliers in the case of permanent increases in labour income taxes. It should be noted, however, that this result does not hold true for all models.

V. Comparison of model-based fiscal multipliers and empirical results

In order to compare our results to those of the VAR studies, which try to deduce the multiplier of government spending either for single euro area countries or (in one case) for the euro area aggregate, some studies' results are comprised in Table 8, based on Boussard et al. (2012) (with the sign convention used in this paper, i.e. the impact of a reduction of government spending by 1% of GDP). The short-term multiplier comprises the impact multiplier and the first four periods, whereas the medium-run multiplier stretches three years into the future. What is striking from those results is the fact that the multipliers vary quite a lot even when focusing on a single country. Short-term multipliers range from -0.23 to -0.7 and long-run multipliers range from -1.27 to 0.23 depending on the sample periods and the identification strategy.

Table 8 around here –

Concerning tax multipliers, Table 9 provides empirical evidence from studies carried out for euro area countries. This table is again largely based on the survey of Boussard et al. (2012). It shows a large heterogeneity among the euro area, with differences even in the sign of the multipliers in the short and in the medium run. Still, except for Germany in Bénassy-Quéré and Cimadomo (2012), empirical estimates reveal that tax shocks usually entail smaller effects on GDP than public expenditure.⁸

- Table 9 around here -

⁸ In another strand of the empirical literature, which is not based on VAR but on simple regression analysis, some authors find on the contrary that fiscal stimulus based on tax cuts are more likely to increase growth than those based on spending increases (see for example Alesina and Ardagna, 2010).

Comparing fiscal multipliers from structural models with those from VAR-based approaches is not straightforward given that the specification of the fiscal shock matters significantly for the results. Furthermore, movements in expenditures and especially in tax revenues in empirical applications may consist of different expenditure and revenue items, making a direct comparison to our model-based results difficult.

Yet another problem is that the main objection raised against the use of SVAR models for the analysis of fiscal policy concerns the identification scheme and the treatment of expectations. Ramey (2011a) shows that government spending shocks identified following a Blanchard and Perotti procedure on U.S. data can be biased due to the fact that spending shocks are not entirely exogenous. In this case, estimated parameters and implied impulse-response functions (IRF) are inconsistent.

In order to address this issue, Ramey (2011a) embeds one of her news variables into a standard SVAR used for the analysis of fiscal policy. It turns out that private consumption decreases and private investment increases on impact for a Blanchard-Perotti contractionary shock, but that the opposite result holds for unanticipated government spending shocks. A recent controversy (Ramey, 2011b, Perotti, 2011) highlighted that these results were sensitive to the inclusion of particular observations and that IRFs for private consumption and GDP in the two cases were probably not significantly different. Using European data, Beetsma and Giuliodori (2011) do not find significant evidence of an anticipation effect either.

Nevertheless, with these caveats in mind, Table 10 presents a rough comparison of our results from structural models to empirical ones. Specifically, we contrast the estimates from the empirical literature with our short-run multipliers for transitory government expenditure and tax shocks.

- Table 10 around here -

Table 10 shows that the empirical VAR-based literature typically finds a large range of short-run fiscal multipliers for both expenditures and tax items. The most important finding is perhaps the fact that in the empirical applications, multipliers can be substantially larger than one in absolute terms for both expenditures and tax items. Another important difference is that some empirical applications find a positive tax multiplier, pointing towards the so called non-Keynesian effects of tax shocks. In the structural models used here, positive tax multipliers are found only in a few cases, e.g. when the consumption tax shock is deemed permanent and labour income tax adjusts in the long run (e.g. Italy and Estonia, see Table 7). In this case, the non-Keynesian effect arises at least partly due to the anticipation of long-run gains of lower labour income taxes in the future.

VI. Conclusions

We have provided estimates of the size and sign of fiscal multipliers - both in the short and in the long run - for European countries based on simulations of structural models used at the NCBs and the ECB. The heterogeneity of the models with regard to the specific model features and the calibration of parameters provided a useful environment to study the driving factors of fiscal multipliers. Differences in the size of the various fiscal multipliers can be traced back to the different nature of the fiscal shocks, as well as some country specific features, such as the share of liquidity-constrained consumers, financial frictions, and different degrees of price and wage rigidities. At the same time, while acknowledging the importance of these country differences, some of the findings are fairly robust across the variety of models.

Our first robust result is that under standard monetary policy short-run multipliers are smaller than one in absolute terms in the vast majority of models, irrespective of the fiscal instrument, the considered country or the nature of the fiscal shock. Temporary reductions in government consumption are typically associated with larger short-run GDP effects than temporary increases in the households' labour income tax rate, the capital income tax rate, or the consumption tax rate.

The second robust finding is that a two-year long ZLB episode has relatively small effects on the multipliers if a temporary fiscal shock hits an individual country within the euro area. The reason is that cross-country spillovers turn out to be rather weak and the response of inflation to the country-specific fiscal shocks is in general rather muted. In contrast, when the fiscal shocks are simultaneously implemented in the euro area as a whole, the ZLB has a relatively strong impact on short-run government consumption multipliers, which can become larger than one. The same holds true for non-euro area countries that exhibit a country-specific monetary policy rule.

Third, if fiscal shocks are implemented permanently, short-run government consumption and consumption tax multipliers are smaller in absolute value than in case of a temporary implementation. Long-run multipliers are in general negative when the budgetary room materialising after the fiscal tightening is used to adjust lump-sum taxes. Instead, long-run multipliers are typically positive if the households' labour income tax rate is reduced in the medium to long term. Since households anticipate these long-run GDP effects at the outset of the simulations, short-run multipliers are more favourable when the budgetary room that materialises after the fiscal tightening is used to lower distortionary taxes.

Overall, our results and the review of empirical literature suggest that many factors affect the size of multipliers. As such, short and long-run multipliers can give rather different policy conclusions as regards the desirability of using a particular fiscal instrument. Expenditure-based fiscal adjustments typically have larger negative short-run effects than tax-based adjustments. However, in the long run, tax-based fiscal adjustments lower the long-run output potential of the economy, while expenditure-based fiscal adjustments can result in positive long-run output effects.

References

Adolfson, M., S. Laséen, L. Christiano, M. Trabandt and K. Walentin (2013). "Ramses II – Model Description," Sveriges Riksbank Occasional Paper Series 12.

Afonso, A. and R. M. Sousa (2009). "The macroeconomic effects of fiscal policy in Portugal: a Bayesian SVAR analysis," School of Economics and Management, Working Papers No. 09/2009/DE/UECE.

Alesina A. and S. Ardagna (2010). "Large Changes in Fiscal Policy: Taxes versus Spending," in Brown R., NBER book *Tax Policy and the Economy*, Vol. 24, pp. 35-68.

Almeida V., G. Castro, R.M. Félix, P. Júlio and J.R. Maria (2013). "Inside PESSOA - A detailed description of the model," Banco de Portugal Working Papers No. 16/2013.

Almeida, V., Castro, G., Félix, R.M. and Maria, J. (2013). "Fiscal Consolidation in a Small Euro-Area Economy," *International Journal of Central Banking*, Vol. 9(4), pp. 1-38.

Ambrisko, R., J. Babecky, J. Rysanek and V. Valenta (2012). "Assessing the Impact of Fiscal Measures on the Czech Economy," Working Papers 2012/15, Czech National Bank.

Andrle, M., T. Hledik, O. Kamenik, and J. Vlcek (2009). "Implementing the New Structural Model of the Czech National Bank." Working Papers 2009/2, Czech National Bank.

Auerbach, A. and Y. Gorodnichenko (2012). "Measuring the Output Responses to Fiscal Policy," *American Economic Journal: Economic Policy*, Vol. 4(2), pp. 1-27.

Auerbach A.J. and Y. Gorodnichenko (2013). "Output Spillovers from Fiscal Policy," *American Economic Review*, Vol. 103(3), pp. 141-46.

Barro, R. J. and C. J. Redlick (2009). "Macroeconomic Effects from Government Purchases and Taxes," NBER Working Paper 15369.

Baum, A, Köster, G. (2011). "The impact of fiscal policy on economic activity over the business cycle-evidence from a threshold VAR analysis," Bundesbank Discussion Paper No. 03/2011

Baum, A., Poplawski-Ribeiro, M., Weber, A. (2012). "Fiscal Multipliers and the State of the Economy," IMF Working Paper.

Beetsma, R. and M. Giuliodori (2011). "The Effects of Government Purchases Shocks: Review and Estimates for the EU," *The Economic Journal*, Vol. 121, pp. 4-32.

Beetsma, R., Giuliodori, M. and F. Klaassen (2006). "Trade spillovers of fiscal policy in the european union: a panel analysis," *Economic Policy*, Vol. 21(48), pp. 639-687.

Bénassy-Quéré, A. and J. Cimadomo (2012). "Changing patterns of fiscal policy multipliers in Germany, the UK and the US," *Journal of Macroeconomics*, Vol. 34(3), pp. 845-873.

Benes, J., T. Hlédik, M. Kumhof and D. Vávra (2005). "An Economy in Transition and DSGE: What are the Czech National Bank's New Projection Model Needs," CNB Working Paper No. 12/2005.

Bernanke, B., Gertler, M. and Gilchrist, S. (1999), "The Financial Accelerator in a Quantitative Business Cycle Framework", in: J. B. Taylor & M. Woodford (ed.), Handbook of Macroeconomics, edition 1, volume 1, chapter 21, pages 1341-1393 Elsevier.

Biau, O. and E. Girard (2005), "Politique budgétaire et dynamique économique en France: l'approche VAR structurel," *Revue économique*, Vol. 56, pp. 755-764

Blanchard, O. and R. Perotti (2002). "An Empirical Characterization of the Dynamic Effects of Changes in Government Spending and Taxes on Output," *The Quarterly Journal of Economics* 117 (4), 1329–1368.

Bouakez H., D. Larocque and M. Normandin (2013). "Separating the Wheat from the Chaff: A Disaggregate Analysis of the Effects of Public Spending in the U.S." Mimeo.

Boussard, J., de Castro, F. and Salto, M. (2012). "Fiscal Multipliers and Public Debt Dynamics in Consolidations," European Economy, Economic Papers 460, July 2012.

Burriel, P., de Castro, F., Garrote, D., Gordo, E., Paredes, J. and Pérez, J.J. (2010). "Fiscal policy shocks in theeuro area and the US: an empirical assessment," *Fiscal Studies* 31(2), 251–285.

Canova, F., M. Ciccarelli and P. Dallari (2013). "Spillover of fiscal shocks in the euro area." Mimeo.

Carlstrom, C., T. Fuerst, and M. Paustian (2012). "Policy multipliers under an interest rate peg of deterministic versus stochastic duration," Bank of England Working Paper No. 475.

Christiano C., M. Eichenbaum and S. Rebelo (2011). "When is the Government Spending Multiplier Large?" *The Journal of Political Economy*, Vol 119(1), pp. 78-121.

Cléaud G., M. Lemoine and P.-A. Pionnier (2013). "Which size and evolution of the government expenditure multiplier in France (1980-2010)?" Mimeo.

Coenen, G., C. Erceg, C. Freedman, D. Furceri, M. Kumhof, R. Lalonde, D Laxton, J. Lindé, A. Mourougane, D. Muir, S. Mursula, C. de Resende, J. Roberts, W. Roeger, S. Snudden, M. Trabandt, and J. in't Veld (2012). "Effects of Fiscal Stimulus in Structural Models," *American Economic Journal: Macroeconomics*, 4(1): 22–68.

Coenen, G., P. McAdam and R. Straub (2008). "Tax reform and labour-market performance in the euro area: A simulation-based analysis using the New Area-Wide Model," *Journal of Economic Dynamics and Control*, Vol. 32, pp. 2543-2583.

Cogan, J., T. Cwik, J. Taylor and V. Wieland (2010). "New Keynesian versus old Keynesian government spending multipliers," *Journal of Economic Dynamics and Control*, 34(3), pp. 281-295.

Cwik, T. and V. Wieland (2011). "Keynesian government spending multipliers and spillovers in the euro area," *Economic Policy*, July 2011, pp. 493-549.

de Castro, F. (2006). "The macroeconomic effects of fiscal policy in Spain," *Applied Economics*, 38, 913-924.

de Castro, F and L. Fernández (2011). "The Effects of Fiscal Shocks on the Exchange Rate in Spain," Banco de España Working Paper No. 1121.

de Castro, F and P. Hernández de Cos (2008). "The economic effects of fiscal policy: the case of Spain," *Journal of Macroeconomics*, 30, 1005-1028.

De Nederlandsche Bank (2011). "DELFI DNB's Macroeconomic Policy Model of the Netherlands," Occasional Studies Vol. 9, No. 1.

Eggertsson, G. (2011). "What fiscal policy is effective at zero interest rates?" NBER Macroeconomics Annual 2010, 25, pp. 59-112.

Forni, L., A. Gerali and M. Pisani (2010). "The macroeconomics of fiscal consolidations in a monetary union: the case of Italy," Banca d'Italia Working Paper 747.

Gadatsch, Hauzenberger, Stähler (2014). Unpublished manuscript.

- Gelain, P. and D. Kulikov (2009). "An Estimated Dynamic Stochastic General Equilibrium Model for Estonia," Eesti Pank Working Paper No. 5/2009.
- Giordano, R., S. Momigliano, S. Neri and R. Perotti (2007). "The effects of fiscal policy in Italy: Evidence from a VAR model," *European Journal of Political Economy*, 23, pp. 707–733.
- Gomes, S., P. Jacquinot and M. Pisani (2012). "The EAGLE. A model for policy analysis of macroeconomic interdependence in the euro area," Economic Modelling, Elsevier, vol. 29(5), pages 1686-1714.
- Hall, R. E. (2009). "By How Much Does GDP Rise If the Government Buys More Output?" *Brookings Papers on Economic Activity*, Fall 2009, pp. 183–231.
- Heppke-Falk, K.H., J. Tenhofen and G.B. Wolff (2006). "The macroeconomic effects of exogenous fiscal policy shocks in Germany: a disaggregated SVAR analysis," *Journal of Economics and Statistics (Jahrbücher für Nationalökonomie und Statistik)*, Vol. 230(3), pp. 328-255.
- in 't Veld (2013). "Fiscal consolidations and spillovers in the Euro area periphery and core," European Economy, Economic Papers 506, October 2013.
- Jacquinot P. and M. Lemoine (2013). "Quantifying the economic impact of fiscal harmonization between France and Germany," mimeo.
- Jemec, N., A. Kastelec and A. Delakorda (2013). "How do fiscal shocks affect the macroeconomic dynamics in the Slovenian economy?" Bank of Slovenia Working Paper 1/2013.
- Kilponen, J., H. Kinnunen and A. Ripatti (2006). "Population ageing in a small open economy some policy experiments with a tractable general equilibrium model," Bank of Finland Research Discussion Papers 28/2006.
- Kirchner, M., J. Cimadomo, and S. Hauptmeier (2010). "Transmission of Government Spending Shocks in the Euro Area Time Variation and Driving Forces," ECB Working Paper No. 1219.
- Laséen, S., and L.E.O. Svensson (2011). "Anticipated Alternative Policy Rate Paths in Policy Simulations," *International Journal of Central Banking*, Vol. 7(3), pp. 1-35.
- Locarno, A., A. Notarpietro and M. Pisani (2013). "Sovereign risk, monetary policy and fiscal multipliers: a structural model-based assessment," Bank of Italy Working Paper No. 943.
- Micallef, B. (2013). "Measuring the Effects of Structural Reforms in Malta: An Analysis using the EAGLE model," Central Bank of Malta Working Paper No. 1/2013.
- Papageorgiou, D. (2014). "BoGGEM: A Dynamic Stochastic General Equilibrium Model for Policy Simulations," Bank of Greece Working Paper No. 182.

Perotti, R. (2004). "Estimating the effects of fiscal policy in OECD countries," Proceedings, Federal Reserve Bank of San Francisco.

Perotti, R. (2011). "Expectations and Fiscal Policy: An Empirical Investigation," mimeo.

Ramey, V. A. (2011a). A Reply to Roberto Perotti's "Expectations and Fiscal Policy: An Empirical Investigation". Mimeo.

Ramey, V. A. (2011b). "Identifying Government Spending Shocks: It's All in the Timing," *The Quarterly Journal of Economics*, Vol. 126(1), pp. 1–50.

Staehler, N., and C. Thomas (2012). "FiMod – A DSGE model for fiscal policy simulations," *Economic Modelling*, Vol. 29(2), pp. 239-261.

Woodford, M. (2011). "Simple Analytics of the Government Expenditure Multiplier," *American Economic Journal: Macroeconomics*, 3(1), pp. 1-35.

Table 1. Short-run fiscal multipliers: Temporary reduction in government consumption

	No ZLB		2-year ZLB	
	Year 1	Year 2	Year 1	Year 2
Belgium	-0.93	-0.90	-0.97	-0.95
Czech Republic	-0.54	-0.54	-1.79	-1.57
Estonia	-0.83	-0.66	-0.98	-0.77
Euro area	-0.98	-0.91	-1.39	-1.30
Finland*	-0.78	-0.76	-0.78	-0.76
France	-0.92	-0.71	-1.05	-0.87
Germany	-0.52	-0.48	-0.72	-0.68
Greece*	-0.90	-0.73	-0.90	-0.73
Italy	-0.79	-0.67	-0.86	-0.73
Malta	-0.73	-0.49	-0.73	-0.49
Netherlands*	-0.74	-0.72	-0.74	-0.72
Portugal*	-0.76	-0.23	-0.76	-0.23
Portugal* (ff)	-0.85	-0.37	-0.85	-0.37
Slovenia	-0.66	-0.48	-0.68	-0.50
Spain	-0.50	-0.29	-0.50	-0.29
Sweden	-0.60	-0.63	-1.63	-2.07

^{*} In these countries, monetary policy is exogenous. Portugal (ff) indicates the presence of financial frictions following Bernanke, Gertler and Gilchrist (1999).

Table 2. Short-run fiscal multipliers: Temporary increase in households' labour income tax rate

	No ZLB		2 year ZLB	
	Year 1	Year 2	Year 1	Year 2
Belgium	-0.04	-0.10	-0.03	-0.10
Czech Republic	-0.36	-0.40	-0.38	-0.28
Estonia	-0.21	-0.43	0.04	-0.22
Euro area	-0.11	-0.19	-0.04	-0.12
Finland*	-0.10	-0.13	-0.10	-0.13
France	-0.13	-0.30	-0.09	-0.25
Germany	-0.10	-0.09	-0.15	-0.14
Greece*	-0.50	-0.77	-0.50	-0.77
Italy	-0.06	-0.13	-0.05	-0.12
Malta	-0.09	-0.20	-0.09	-0.20
Netherlands*	-0.11	-0.15	-0.11	-0.15
Portugal*	-0.51	-0.91	-0.51	-0.91
Portugal* (ff)	-0.49	-0.86	-0.49	-0.86
Slovenia	-0.10	-0.19	-0.10	-0.19
Spain	-0.13	-0.11	-0.13	-0.11
Sweden	-0.27	-0.31	0.56	0.88

^{*} In these countries, monetary policy is exogenous. Portugal (ff) indicates the presence of financial frictions following Bernanke, Gertler and Gilchrist (1999).

Table 3. Short-run fiscal multipliers: Temporary increase in capital tax rate

	No ZLB		2 year ZLB	
	Year 1	Year 2	Year 1	Year 2
Belgium	-0.06	-0.08	-0.06	-0.08
Estonia	-0.10	-0.11	-0.10	-0.12
Euro area	-0.12	-0.10	-0.19	-0.17
Finland*	-0.10	-0.12	-0.10	-0.12
France	-0.07	-0.08	-0.09	-0.10
Germany	-0.05	-0.08	-0.11	-0.14
Greece*	-0.65	-1.06	-0.65	-1.06
Italy	-0.08	-0.11	-0.09	-0.12
Malta	-0.02	-0.04	-0.02	-0.04
Portugal*	-0.10	-0.01	-0.10	-0.01
Portugal* (ff)	-0.19	-0.15	-0.19	-0.15
Slovenia	-0.11	-0.11	-0.12	-0.12
Spain	-0.09	-0.07	-0.09	-0.07
Sweden	-0.33	-0.50	-2.18	-3.14

^{*} In these countries, monetary policy is exogenous. Portugal (ff) indicates the presence of financial frictions following Bernanke, Gertler and Gilchrist (1999).

Table 4. Short-run fiscal multipliers: Temporary increase in consumption tax rate

	No.	ZLB	2 yea	r ZLB
	Year 1	Year 2	Year 1	Year 2
Belgium	-0.19	-0.43	-0.20	-0.43
Czech Republic	-0.19	-0.09	-0.15	-0.03
Estonia	-0.25	-0.08	-0.25	-0.08
Euro area	-0.48	-0.62	-0.78	-0.92
Finland*	-0.72	-0.70	-0.72	-0.70
France	-0.14	-0.23	-0.18	-0.29
Germany	-0.17	-0.22	-0.17	-0.17
Greece*	-0.48	-0.56	-0.48	-0.56
Italy	-0.29	-0.36	-0.35	-0.41
Malta	-0.15	-0.18	-0.15	-0.18
Portugal*	-0.49	-0.38	-0.49	-0.38
Portugal* (ff)	-0.52	-0.43	-0.52	-0.43
Slovenia	-0.24	-0.25	-0.24	-0.25
Spain	-0.14	-0.19	-0.14	-0.19
Sweden	-0.17	-0.21	-1.05	-1.45

^{*} In these countries, monetary policy is exogenous. * Portugal (ff) indicates the presence of financial frictions following Bernanke, Gertler and Gilchrist (1999).

Table 5. Short and long-run fiscal multipliers: Permanent reduction in government consumption

Fiscal rule:	lump-sum tax			households' labour income tax			
	Year 1	Year 2	long run	Year 1	Year 2	long run	
Belgium	-0.95	-0.90	-0.63	-0.93	-0.83	0.70	
Czech Republic	-0.25	-0.21	-0.43				
Euro area	-0.83	-0.62	-0.61	-0.46	-0.29	0.34	
Estonia	-0.65	-0.61	-0.68	-0.32	-0.22	0.84	
Finland*	-0.40	-0.31	-0.63	-0.33	-0.25	0.91	
France	-0.97	-0.76	-0.82	-0.82	-0.48	1.28	
Germany	-0.62	-0.40	-0.24	-0.61	-0.51	0.06	
Greece*	-0.87	-0.74	-1.05	-0.83	-0.81	0.53	
Italy	-0.68	-0.52	-0.58	-0.51	-0.19	0.54	
Malta	-0.68	-0.37	-0.51	-0.62	-0.21	0.30	
Portugal*	-0.58	-0.35	-0.67	-0.62	-0.05	1.64	
Portugal* (ff)	-0.67	-0.44	-0.66	-0.72	-0.20	1.55	
Slovenia	-0.66	-0.41	-0.38	-0.56	-0.15	0.82	
Spain	-0.57	-0.35	-0.39	-0.48	-0.38	0.31	
Sweden	-0.48	-0.44	-0.60				

^{*} In these countries, monetary policy is exogenous. Portugal (ff) indicates the presence of financial frictions following Bernanke, Gertler and Gilchrist (1999).

Table 6. Short and long-run fiscal multipliers: Permanent increase in tax rate - lump-sum taxes adjust

Tax rate:	labou	ır incon	ne tax	capita	ıl incom	e tax	consumption tax		
	Year 1	Year 2	long run	Year 1	Year 2	long run	Year 1	Year 2	long run
Belgium	-0.02	-0.18	-1.03	-0.29	-0.58	-2.11	-0.24	-0.49	-0.57
Czech Republic	-0.20	-0.32	-0.11				-0.03	-0.07	-0.03
Euro area	-0.52	-0.66	-0.87	-1.69	-2.21	-2.56	-0.40	-0.45	-0.51
Estonia	-0.56	-0.65	-0.60	-0.92	-0.76	-1.25	0.00	0.01	-0.16
Finland*	-0.79	-0.64	-1.48	-0.12	-0.99	-1.97	-0.47	-0.10	-0.74
France	-0.28	-0.63	-1.24	-0.36	-0.61	-3.27	-0.18	-0.36	-0.61
Germany	-0.19	-0.15	-0.29	-0.02	-0.11	-0.79	-0.04	-0.06	-0.13
Greece*	-0.57	-0.82	-1.41	-1.18	-2.46	-3.77	-0.39	-0.58	-0.96
Italy	-0.19	-0.38	-0.91	-0.21	-0.57	-2.50	-0.08	-0.15	-0.36
Malta	-0.14	-0.33	-0.72	-0.06	-0.16	-1.67	-0.09	-0.17	-0.31
Portugal*	-0.47	-1.04	-1.27	-0.34	-0.45	-2.01	-0.29	-0.52	-0.66
Portugal* (ff)	-0.45	-0.98	-1.27	-0.54	-0.53	-2.00	-0.28	-0.50	-0.66
Slovenia	-0.26	-0.55	-1.42	-0.48	-0.77	-3.26	-0.13	-0.23	-0.54
Spain	-0.12	-0.11	-0.53	-0.26	-0.45	-3.25	-0.16	-0.18	0.00
Sweden	-0.35	-0.50	-0.68	-0.43	-0.80	-1.81	-0.15	-0.21	-0.28

^{*} In these countries, monetary policy is exogenous. Lump-sum transfers adjust in the long run. Portugal (ff) indicates the presence of financial frictions following Bernanke, Gertler and Gilchrist (1999).

Table 7. Short and long-run fiscal multipliers: Permanent increase in tax rate – households' labour income tax rate adjusts

Tax rate:	capital income tax			consumption tax		
	Year 1	Year 2	long run	Year 1	Year 2	long run
Belgium	-0.29	-0.44	-1.04	-0.18	-0.35	0.53
Czech Republic						
Euro area	-1.23	-1.82	-1.17	-0.09	-0.17	0.33
Estonia	-0.48	-0.64	-0.16	0.27	0.38	1.73
Finland*	0.13	-0.91	-1.52	-0.37	-0.28	1.07
France	-0.22	-0.41	-2.43	-0.05	-0.11	1.31
Germany	-0.14	-0.15	-0.98	-0.17	-0.20	1.41
Greece*	-1.17	-2.51	-2.69	-0.35	-0.56	0.55
Italy	-0.08	-0.30	-1.92	0.10	0.20	0.66
Malta	-0.02	-0.08	-1.26	-0.02	0.01	0.47
Portugal*	-0.34	-0.17	-0.79	-0.30	-0.36	0.58
Portugal* (ff)	-0.57	-0.30	-0.82	-0.31	-0.37	0.53
Slovenia	-0.39	-0.52	-2.36	-0.02	0.07	0.59
Spain	-0.29	-0.48	-2.79	-0.18	-0.21	0.74

^{*} In these countries, monetary policy is exogenous. Labour income tax rate adjusts in the long run. Portugal (ff) indicates the presence of financial frictions following Bernanke, Gertler and Gilchrist (1999).

Table 8. VAR-based fiscal multipliers: reduction in government spending

Studies	Sample (Country and Periods)	Short-term Multiplier	Medium- term Multiplier
Perotti (2004)	Germany (1960:1-1974:4)	-0.53	0.27
	Germany (1975:1-1989:4)	-0.50	-0.07
Heppke-Falk et al. (2006)	Germany (1974:1–2004:4)	-0.62	-1.27
Baum and Koester (2011)	Germany (1976:1-2009:4)	-0.7	-0.69
Bénassy-Quéré and Cimadomo (2012)	Germany (1971:1-2004:4)	-0.23	0.23
Biau and Girard (2005)	France (1978:1-2003:4)	-1.4	not significant
Cléaud et al. (2013)	France (1980:1-2010:4)	-1.1	-0.5
Giordano et al. (2007)	Italy (1982:1-2004:4)	-1.2	-1.7
de Castro (2006)	Spain (1980:1-2001:2)	-1.14 to -1.54	-0.58 to -1.04
de Castro and Hernàndez de Cos (2008)	Spain (1980:1-2004:4)	-1.3	-1
de Castro and Fernández (2011)	Spain (1981:1-2008:4)	-0.94	-0.55
Jemec et al. (2013)	Slovenia (1995:1-2010:4)	-1.6	not significant
Burriel et al. (2010)	Euro Area (1981:1-2007:4)	-0.87	-0.85
Kirchner et al. (2010)	Euro Area (1980:1-2008:4)	-0.7 to -1.0	0.7 to 1.7

Table 9. VAR-based fiscal multipliers: Increase in net taxes

Studies	Sample (country and period)	Short-term Multiplier	Medium-term Multiplier
Dorotti (2004)	Germany (1960:1-1974:4)	0.29	-0.05
Perotti (2004)	Germany (1975:1-1989:4)	-0.04	0.59
Baum and Koester (2011)	Germany (1976:1-2009:4)	-0.66	-0.53
Bénassy-Quéré and Cimadomo (2006)	Germany (1971:1-2004:4)	-1.17	-1.08
Biau and Girard (2005)	France (1978:1-2003:4)	-0.1	not significant
Giordano et al. (2007)	Italy (1982:1-2004:4)	0.16	
De Castro (2006)	Spain (1980:1-2001:2)	0.05	0.39
Afonso and Sousa (2009)	Portugal (1979:1-2007:4)	+	+
Jemec et al. (2013)	Slovenia (1995:1-2010:4)	-0.4	not significant
Burriel et al. (2010)	Euro Area (1981-2007)	-0.63	-0.49

Table 10. Comparing fiscal multipliers in structural and empirical models

	Empirical*		Structura	Models**	
	Min	Max	Min	Max	
Expenditure Multipliers	-1.54	-0.23	-0.98	-0.50	
Tax Multipliers	-1.17	0.29	-0.72	-0.02	
- Labour			-0.51	-0.02	
- Consumption			-0.72	-0.15	
- Capital			-0.65	-0.02	

^{*} Minimum and maximum values are based on Tables 1-2, first column (Short Run)

 $^{^{**}}$ Minimum and maximum values are based on Tables 3-6, first column (Year 1, NO-ZLB)

Appendix

Table A1. Models used in the simulation exercises

Country	Model	Reference
Belgium	na	
Czech Republic	g3	Ambrisko, Babecky, Rysanek and Valenta (2012)
Estonia	EP DSGE	Gelain and Kulikov (2009)
Euro area	NAWM	Coenen, McAdam and Straub (2008)
Finland	Aino	Kilponen, Kinnunen and Ripatti (2006)
France	EAGLE	Jacquinot and Lemoine (2013)
Germany	GEAR	Gadatsch, Hauzenberger and Stähler (forthcoming, 2014)
Greece	BoGGEM	Papageorgiou (forthcoming, 2014)
Italy	IDEA-BI-EAGLE	Forni, Gerali and Pisani (2010)
Malta	EAGLE	Micallef (2013)
Netherlands	DELFI	De Nederlandsche Bank (2011)
Portugal	PESSOA	Almeida, Castro, Félix, Júlio and Maria (2013)
Slovenia	EAGLE	Gomes, Jacquinot and Pisani (2010)
Spain	FiMod	Stähler and Thomas (2012)
Sweden	Ramses II	Adolfson, Laséen, Christiano, Trabandt, Walentin (2013)

Table A2. Elements of calibration

	Belg.	Cz. Rep.	Estonia	EA	Finland
Name of the model		g3	EP DSGE	NAWM	Aino
Model calibrated/estimated	est.	est.	est.	cal.	cal.
Open economy features					
Number of countries	3	2	2	2	1
Number of countries in monetary Union	2	0	1	1	1
RoW/RoEA exogenous	no	yes	yes	no	yes
Tradable/nontradable goods	both	tr. only	both	tr.	tr.only
Steady state values					
Private consumption-to-GDP ratio	0.60	0.59	0.60	0.60	0.62
Private investment-to-GDP ratio	0.15	0.12	0.25	0.22	0.19
Imports-to-GDP ratio	0.74	0.29	0.90	0.18	0.38
Public consumption-to-GDP ratio	0.14	0.22	0.25	0.16	0.17
Public investment-to-GDP ratio	0.00	0.06	0.00	0.03	0.02
Public sector interest payment-to-GDP ratio	0.03	0.01	0.00	0.05	0.03
Labour income tax revenues-to-GDP ratio	0.31	0.11	0.11	0.07	0.14
Capital income tax revenues-to-GDP ratio	0.04	0.01	0.03	0.03	0.03
Consumption tax revenues-to-GDP ratio	0.10	0.12	0.11	0.11	0.14
Value of the public debt-to-annualized GDP	0.60	0.45		0.90	0.60
Value of the net foreign asset-to-yearly GDP	0.00	0.00		0.00	0.00
Annualized nominal interest rate	0.05	0.03	0.05	0.05	0.05
Annualized inflation	0.02	0.02	0.00	0.02	0.02
Calibration					
Share of liquidity-constrained households	0.00	0.40	0.00	0.25	0.00
Coefficient of risk aversion	2.12	N/A	1.61	1.00	3.00
Frisch elasticity of labour supply	2.08	2.84	1.78	2.00	> 2
Habit	0.66	0.75	0.65	0.60	0.00
Adjustment costs on investment	13.66	0.20	6.42	3.00	1.40
Price stickiness	0.71	0.50	0.69	0.90	0.85
Price indexation	0.59		0.27	0.70	1.00
Wage stickiness	0.78	0.80	0.55	0.75	0.85
Wage indexation	0.90		0.37	0.75	1.00
Own Taylor rule (cal. as ECBWP1195)	no	yes	no	yes	no
Fiscal rule react on deviation of pub. debt	yes	yes	yes	no	yes
Fiscal rule react on deviation of pub. deficit	no	no	no	yes	no
Fiscal rule react on deviation of gov. cons.	no	yes	no	no	no

Table A3. Elements of calibration

	France	Germany	Greece	Italy	Malta
Name of the model	EAGLE			EAGLE	EAGLE
Model calibrated/estimated	cal.	cal.	cal.	cal.	cal.
Open economy features					
Number of countries	5	3	1	3	4
Number of countries in monetary Union	3	2	1	2	2
RoW/RoEA exogenous	no	no	yes	no	no
Tradable/nontradable goods	both		tr. only	both	both
Steady state values					
Private consumption-to-GDP ratio	0.57	0.62	0.63	0.59	0.63
Private investment-to-GDP ratio	0.19	0.23	0.26	0.18	0.20
Imports-to-GDP ratio	0.27	0.22	0.17	0.25	0.50
Public consumption-to-GDP ratio	0.23	0.12	0.18	0.20	0.20
Public investment-to-GDP ratio	0.00	0.03	0.03	0.02	0.00
Public sector interest payment-to-GDP ratio	0.03	0.01	0.05	0.04	0.03
Labour income tax revenues-to-GDP ratio	0.24	0.35	0.23	0.21	0.15
Capital income tax revenues-to-GDP ratio	0.05	0.02	0.09	0.13	0.03
Consumption tax revenues-to-GDP ratio	0.10	0.09	0.11	0.10	0.11
Value of the public debt-to-annualized GDP	0.62	0.60	1.20	1.19	0.60
Value of the net foreign asset-to-yearly GDP	0.04	0.00	0.00	0.00	0.04
Annualized nominal interest rate	0.05	0.02	0.04	0.03	0.05
Annualized inflation	0.02	1.80	0.00	0.00	0.02
Calibration					
Share of liquidity-constrained households	0.25	0.40	0.40	0.00	0.25
Coefficient of risk aversion	1.00	1.40	1.00	1.00	1.00
Frisch elasticity of labour supply	2.00	11.00	1.00	0.50	2.00
Habit	0.90	0.60	0.65	0.60	0.70
Adjustment costs on investment	6.00	6.90	10.00	6.00	4.00
Price stickiness	0.75	0.90	0.71	0.75-0.8	0.75
Price indexation	0.75	0.45	0.27	0.50	0.50
Wage stickiness	0.92	200.00	BG07	0.75-0.8	0.75
Wage indexation	0.50	0.75		0.50	0.75
Own Taylor rule (cal. as ECBWP1195)	no	no	no	no	no
Fiscal rule react on deviation of pub. debt	yes	yes	yes	yes	yes
Fiscal rule react on deviation of pub. deficit	no	no	no	yes	no
Fiscal rule react on deviation of gov. cons.	no	no	no	no	no

Table A4. Elements of calibration

Table A4. Elements of calibration	Nether*.	Portugal	Slovenia	Spain	Sweden
Name of the model	DELFI	PESSOA	EAGLE	FiMod	Ramses II
Model calibrated/estimated	est.	cal.	cal.	cal.	est.
Open economy features					
Number of countries	1	1	4	2	2
Number of countries in monetary Union	1	1	2	2	0
RoW/RoEA exogenous	yes	yes	no	no	yes
Tradable/nontradable goods		tr. only	both	tr. only	tr. only
Steady state values					
Private consumption-to-GDP ratio		0.60	0.55	0.57	0.63
Private investment-to-GDP ratio		0.21	0.27	0.21	0.17
Imports-to-GDP ratio		0.33	0.61	0.27	0.44
Public consumption-to-GDP ratio		0.23	0.19	0.18	0.30
Public investment-to-GDP ratio				0.04	0.00
Public sector interest payment-to-GDP ratio		0.02	0.03	0.02	0.00
Labour income tax revenues-to-GDP ratio		0.11	0.23	0.07	0.29
Capital income tax revenues-to-GDP ratio		0.03	0.01	0.02	
Consumption tax revenues-to-GDP ratio		0.43	0.09	0.04	0.16
•					
Value of the public debt-to-annualized GDP		0.53	0.60	0.48	0.00
Value of the net foreign asset-to-yearly GDP		-0.23	-0.09	0.00	0.00
Annualized nominal interest rate		0.05	0.05	0.04	0.04
Annualized inflation		0.02	0.02	0.00	0.02
Calibration		0.40	0.05	0.40	0.00
Share of liquidity-constrained households		0.40	0.25	0.40	0.00
Coefficient of risk aversion		5.00	1.00	2.00	1.00
Frisch elasticity of labour supply		0.85	2.00	match. funct.	2.98
Habit		0.90	0.80	0.85	0.66
Adjustment costs on investment		10.00	5.00	2.50	2.35
Price stickiness		100.00	0.75	0.75	0.88
Price indexation		0.00	0.50	0.00	0.16
Wage stickiness		100.00	0.81	0,75/0,7	0.75
Wage indexation		0.00	0.75	0.50	0.34
Own Taylor rule (cal. as ECBWP1195)	no	no	no	no	yes
Fiscal rule react on deviation of pub. debt		yes	yes	yes	no
Fiscal rule react on deviation of pub. deficit		no	no	no	no
Fiscal rule react on deviation of gov. cons.		no	no	no	yes

^{*}Netherland's model is not a DSGE model, hence some of the data is not available.

Table A5. Fiscal multipliers: Sensitivity with respect to key parameters

Panel A. Average multipliers across models in the first two years

Panel A. Average multipliers across models in the first two years							
	Baseline	More				Higher	
		liquidity	Less	Less	Lower	investment	
		constrained	price	wage	risk	adjustment	
		households	stickiness	stickiness	aversion	costs	
-		Households	Stickings	SUCKITICSS	aversion	COSIS	
Government consumption	-0.53	-0.70	-0.50	-0.51	-0.49	-0.54	
Gov.consumption + 2 year ZLB	-0.59	-0.81	-0.59	-0.57	-0.56	-0.60	
Labour income tax	-0.50	-0.56	-0.50	-0.53	-0.51	-0.51	
Labour income tax + 2 year ZLB	-0.55	-0.66	-0.58	-0.59	-0.58	-0.56	
Panel B. Average le	ong-run mu	Itipliers across	models				
Government consumption	-0.56	-0.63	-0.56	-0.56	-0.47	-0.56	
Labour income tax	-1.11	-1.07	-1.12	-1.12	-1.21	-1.12	

Note: The sensitivity analysis is based on the results from the NAWM, 2 multi-country models and 2 small open economy models with fixed interest rate.

NATIONAL BANK OF BELGIUM - WORKING PAPERS SERIES

The Working Papers are available on the website of the Bank: http://www.nbb.be.

- 219. "Comparative advantage, multi-product firms and trade liberalisation: An empirical test", by C. Fuss and L. Zhu, *Research series*, January 2012.
- 220. "Institutions and export dynamics", by L. Araujo, G. Mion and E. Ornelas, Research series, February 2012.
- 221. "Implementation of EU legislation on rail liberalisation in Belgium, France, Germany and the Netherlands", by X. Deville and F. Verduyn, *Document series*, March 2012.
- 222. "Tommaso Padoa-Schioppa and the origins of the euro", by I. Maes, Document series, March 2012.
- 223. "(Not so) easy come, (still) easy go? Footloose multinationals revisited", by P. Blanchard, E. Dhyne, C. Fuss and C. Mathieu, *Research series*, March 2012.
- 224. "Asymmetric information in credit markets, bank leverage cycles and macroeconomic dynamics", by A. Rannenberg, *Research series*, April 2012.
- 225. "Economic importance of the Belgian ports: Flemish maritime ports, Liège port complex and the port of Brussels Report 2010", by C. Mathys, Document series, July 2012.
- 226. "Dissecting the dynamics of the US trade balance in an estimated equilibrium model", by P. Jacob and G. Peersman, *Research series*, August 2012.
- 227. "Regime switches in volatility and correlation of financial institutions", by K. Boudt, J. Daníelsson, S.J. Koopman and A. Lucas, *Research series*, October 2012.
- 228. "Measuring and testing for the systemically important financial institutions", by C. Castro and S. Ferrari, *Research series*, October 2012.
- 229. "Risk, uncertainty and monetary policy", by G. Bekaert, M. Hoerova and M. Lo Duca, *Research series*, October 2012.
- 230. "Flights to safety", by L. Baele, G. Bekaert, K. Inghelbrecht and M. Wei, Research series, October 2012.
- 231. "Macroprudential policy, countercyclical bank capital buffers and credit supply: Evidence from the Spanish dynamic provisioning experiments", by G. Jiménez, S. Ongena, J.-L. Peydró and J. Saurina, *Research series*, October 2012.
- 232. "Bank/sovereign risk spillovers in the European debt crisis", by V. De Bruyckere, M. Gerhardt, G. Schepens and R. Vander Vennet, *Research series*, October 2012.
- 233. "A macroeconomic framework for quantifying systemic risk", by Z. He and A. Krishnamurthy, *Research series*, October 2012.
- 234. "Fiscal policy, banks and the financial crisis", by R. Kollmann, M. Ratto, W. Roeger and J. in't Veld, Research series, October 2012.
- 235. "Endogenous risk in a DSGE model with capital-constrained financial intermediaries", by H. Dewachter and R. Wouters, *Research series*, October 2012.
- 236. "A macroeconomic model with a financial sector", by M.K. Brunnermeier and Y. Sannikov, *Research series*, October 2012.
- 237. "Services versus goods trade: Are they the same?", by A. Ariu, Research series, December 2012.
- 238. "Importers, exporters, and exchange rate disconnect", by M. Amiti, O. Itskhoki and J. Konings, *Research series*, December 2012.
- 239. "Concording EU trade and production data over time", by I. Van Beveren, A.B. Bernard and H. Vandenbussche, *Research series*, December 2012.
- 240. "On the origins of the Triffin dilemma: Empirical business cycle analysis and imperfect competition theory", by I. Maes, *Research series*, December 2012.
- 241. "The Influence of the Taylor rule on US monetary policy", by P. Ilbas, Ø. Røisland and T. Sveen, *Research series*, January 2013.
- 242. "Economic importance of the Belgian ports: Flemish maritime ports, Liège port complex and the port of Brussels Report 2011", by C. Mathys, *Document series*, July 2013.
- 243. "The fragility of two monetary regimes: The European Monetary System and the Eurozone", by P. De Grauwe and Y. Ji, *Research series*, October 2013.
- 244. "Funding liquidity, market liquidity and TED spread: A two-regime model", by K. Boudt, E. C.S. Paulus and D. W.R. Rosenthal, *Research series*, November 2013.
- 245. "Robustifying optimal monetary policy using simple rules as cross-checks", by P. Ilbas, Ø. Røisland and T. Sveen, *Research series*, November 2013.
- 246. "Household and firm leverage, capital flows and monetary policy in a small open economy", by M. Pirovano, *Research series*, November 2013.
- 247. "The BIS and the Latin American debt crisis of the 1980s", by P. Clement and I. Maes, *Research series*, December 2013.
- 248. "The importance of the right amount of business resources for firms' exporting behavior", by I. Paeleman, C. Fuss and T. Vanacker, *Research series*, December 2013.

- 249. "The role of financial frictions during the crisis: An estimated DSGE model", by R. Merola, *Research series*, December 2013.
- 250. "Bank reactions after capital shortfalls", by C. Kok and G. Schepens, Research series, December 2013.
- 251. "Why firms avoid cutting wages: Survey evidence from European firms", by P. Du Caju, T. Kosma, M. Lawless, J. Messina and T. Rõõm, *Research series*, December 2013.
- 252. "The distribution of debt across euro area countries: The role of individual characteristics, institutions and credit conditions", by O. Bover, J. M. Casado, S. Costa, Ph. Du Caju, Y. McCarthy, E. Sierminska, P. Tzamourani, E. Villanueva and T. Zavadil, *Research series*, December 2013.
- 253. "Micro-based evidence of EU competitiveness: The CompNet database", by CompNet Task Force, Research series, March 2014.
- 254. "Information in the yield curve: A macro-finance approach", by H. Dewachter, L. Iania and M. Lyrio, *Research series*, March 2014.
- 255. "The Single supervisory mechanism or 'SSM', part one of the Banking Union", by E. Wymeersch, *Research series*, April 2014.
- 256. "Nowcasting Belgium", by D. de Antonio Liedo, Research series, April 2014.
- 257. "Human capital, firm capabilities and productivity growth", by I. Van Beveren and S. Vanormelingen, Research series, May 2014.
- 258. "Monetary and macroprudential policies in an estimated model with financial intermediation", by P. Gelain and P. Ilbas, *Research series*, May 2014.
- 259. "A macro-financial analysis of the euro area sovereign bond market", by H. Dewachter, L. Iania, M. Lyrio and M. de Sola Perea, *Research series*, June 2014.
- 260. "Economic importance of the Belgian ports: Flemish maritime ports, Liège port complex and the port of Brussels Report 2012", by C. Mathys, *Document series*, June 2014.
- 261. "European competitiveness: A semi-parametric stochastic metafrontier analysis at the firm level", by M. Dumont, B. Merlevede, G. Rayp and M. Verschelde, *Document series*, July 2014.
- 262. "Employment, hours and optimal monetary policy", by M. Dossche, V. Lewis and C. Poilly, *Research series*, September 2014.
- 263. "On the conjugacy of off-line and on-line Sequential Monte Carlo Samplers", by A. Dufays, *Research series*, September 2014.
- 264. "The effects of state aid on Total Factor Productivity growth", by P. Van Cayseele, J. Konings and I. Sergant, *Research series*, October 2014.
- 265. "Assessing the role of ageing, feminising and better-educated workforces on TFP growth", by A. Ariu and V. Vandenberghe, *Research series*, October 2014.
- 266. "A constrained nonparametric regression analysis of factor-biased technical change and TFP growth at the firm level", by M. Verschelde, M. Dumont, B. Merlevede and G. Rayp, *Research series*, October 2014.
- 267. "Market imperfections, skills and total factor productivity: Firm-level evidence on Belgium and the Netherlands", by S. Dobbelaere and M. Vancauteren, *Research series*, October 2014.
- 268. "Import competition, productivity and multi-product firms", by E. Dhyne, A. Petrin, V. Smeets and F. Warzynski, *Research series*, October 2014.
- 269. "International competition and firm performance: Evidence from Belgium", by J. De Loecker, C. Fuss and J. Van Biesebroeck, *Research series*, October 2014.
- 270. "Acquisitions, productivity, and profitability: Evidence from the Japanese cotton spinning industry", by S. Braguinsky, A. Ohyama, T. Okazaki and C. Syverson, *Research series*, October 2014.
- 271. "Total factor productivity: Lessons from the past and directions for the future", by B. van Ark, *Research series*, October 2014.
- 272. "Outward Foreign Direct Investment and domestic performance: In search of a causal link", by E. Dhyne and S. S. Guerin, *Research series*, October 2014.
- 273. "Economic importance of air transport and airport activities in Belgium Report 2012", by F. Van Nieuwenhove, *Document series*, November 2014.
- 274. "Fiscal policy and TFP in the OECD: Measuring direct and indirect effects", by G. Everaert F. Heylen and R. Schoonackers, *Research series*, November 2014.
- 275. "Effectiveness and transmission of the ECB's balance sheet policies", by J. Boeckx, M. Dossche and G. Peersman, *Research series*, December 2014.
- 276. "How do exporters react to changes in cost competitiveness?", by S. Decramer, C. Fuss and J. Konings, *Research series*, January 2015.
- 277. "Optimal monetary policy response to endogenous oil price fluctuations", by A. Stevens, *Research series*, January 2015.
- 278. "Comparing fiscal multipliers across models and countries in Europe", by J. Kilponen, M. Pisani, S. Schmidt, V. Corbo, T. Hledik, J. Hollmayr, S. Hurtado, P. Júlio, D. Kulikov, M. Lemoine, M. Lozej, H. Lundvall, J. R. Maria, B. Micallef, D. Papageorgiou, J. Rysanek, D. Sideris, C. Thomas and G. de Walque, *Research series*, March 2015.

National Bank of Belgium Limited liability company

RLP Brussels - Company's number: 0203.201.340

Registered office: boulevard de Berlaimont 14 – BE-1000 Brussels

www.nbb.be

Editor

Jan Smets

Member of the Board of directors of the National Bank of Belgium

© Illustrations: National Bank of Belgium

Layout: Analysis and Research Group Cover: NBB AG – Prepress & Image

Published in March 2015